

ഹ

Geophysical Research Letters^{*}

RESEARCH LETTER

10.1029/2022GL100191

Key Points:

- Salinization
- Coastal areas
- Groundwater

Supporting Information:

Supporting Information may be found in the online version of this article.

Correspondence to:

G. Nordio, nordiog@bu.edu

Citation:

Nordio, G., Frederiks, R., Hingst, M., Carr, J., Kirwan, M., Gedan, K., et al. (2023). Frequent storm surges affect the groundwater of coastal ecosystems. *Geophysical Research Letters*, *50*, e2022GL100191. https://doi. org/10.1029/2022GL100191

Received 28 JUN 2022 Accepted 5 DEC 2022

Author Contributions:

Conceptualization: Giovanna Nordio, Sergio Fagherazzi Data curation: Giovanna Nordio, Ryan Frederiks, Mary Hingst, Joel Carr Formal analysis: Giovanna Nordio, Sergio Fagherazzi Investigation: Giovanna Nordio Methodology: Giovanna Nordio Supervision: Matt Kirwan, Keryn Gedan, Holly Michael, Sergio Fagherazzi Validation: Sergio Fagherazzi Visualization: Giovanna Nordio Writing – original draft: Giovanna Nordio

Writing – review & editing: Keryn Gedan, Holly Michael, Sergio Fagherazzi

© 2022. The Authors.

This is an open access article under the terms of the Creative Commons Attribution-NonCommercial-NoDerivs License, which permits use and distribution in any medium, provided the original work is properly cited, the use is non-commercial and no modifications or adaptations are made.

Frequent Storm Surges Affect the Groundwater of Coastal Ecosystems

Giovanna Nordio¹, Ryan Frederiks², Mary Hingst², Joel Carr³, Matt Kirwan⁴, Keryn Gedan⁵, Holly Michael², and Sergio Fagherazzi¹

¹Earth and Environment Department, Boston University, Boston, MA, USA, ²Department of Earth Sciences, University of Delaware, Newark, DE, USA, ³Eastern Ecological Science Center, United States Geological Survey, Laurel, MD, USA, ⁴Virginia Institute of Marine Science, College of William and Mary, Gloucester Point, VA, USA, ⁵Department of Biological Sciences, George Washington University, Washington, DC, USA

Abstract Recent studies have focused on the effect of large tropical cyclones (hurricanes) on the shore, neglecting the role of less intense but more frequent events. Here we analyze the effect of the offshore tropical storm Melissa on groundwater data collected along the North America Atlantic coast. Our meta-analysis indicates that both groundwater level and specific conductivity significantly increased during Melissa, respectively reaching maximum values of 1.09 m and 25.2 mS/cm above pre-storm levels. Time to recover to pre-storm levels was 10 times greater for groundwater specific conductivity, with a median value of 20 days, while groundwater level had a median recovery time of 2 days. A frequency-magnitude analysis indicates that the percent of time with salinization is higher for Melissa than for energetic hurricanes. Given the high frequency of these events (return period of 1–2 years), and the long time needed for groundwater conditions to return to normal levels, we conclude that increasingly frequent moderate storms will have a significant impact on the ecology of vegetated shorelines.

Plain Language Summary Salinization and flooding events due to sea level rise and storm surges threaten coastal ecosystems, changing groundwater characteristics. Moderate and more frequent storm surges can have a significant impact on coastal ecology, similar to larger tropical cyclones. Salinity and water table elevation need time to recover to normal conditions. The recovery time is compared to the frequency of these moderate storm surge events to determine the effect on the coastal groundwater.

1. Introduction

Along the coast, salinization and flooding caused by sea level rise (SLR) and storm surges result in the radical conversion of ecosystems, in response to a host of new hydrological, geomorphological, ecological, and biogeochemical dynamics (Tully et al., 2019). Saltwater intrusion and flooding events kill mature trees and suppress germination and seedling survival in forested areas, encouraging their retreat (Antonellini & Mollema, 2010; Fagherazzi, Anisfeld, et al., 2019; Fagherazzi, Nordio, et al., 2019; Kirwan & Gedan, 2019; Munns & Tester, 2008; Pezeshki, 1992; Schieder & Kirwan, 2019; Williams et al., 1999). These events also determine the gradual or sudden decline of croplands (Tully et al., 2019; Williams et al., 1999) and can slow down the leaf growth of less tolerant tree species in coastal urban green spaces (Hallett et al., 2018).

Lateral saltwater intrusion, where saltwater migrates inland through subsurface pathways, is considered one of the most impactful drivers for the salinization of most coastal areas over the long term (Barlow & Reichard, 2009). Though less widely studied, storm surge events, despite occurring on short temporal scales, can have greater impact through vertical saltwater intrusion, caused by flooding and rapid vertical infiltration of saltwater. Recovery time (RT) (the duration of time between initial storm-surge inundation to the return of the groundwater specific conductivity to baseline conditions) and penetration depth (maximum depth reached during vertical infiltration) are dependent on aquifer characteristics (Yang et al., 2018). Anderson (2002) estimated an aquifer recovery period greater than 3 years for a major hurricane overwash in a barrier island. The RT of the coastal aquifer in the Pukapuka Atoll after the category 5 cyclone Percy in 2005 was estimated to be around 1 year (Terry & Falkland, 2010). A similar event, occurred in a lower-permeability surficial aquifer, was felt up to 8 years after (Xiao et al., 2019). In 2013, saltwater intrusion due to supertyphoon Haiyan contaminated sandy aquifers of Samar Island, Philippines. The RT was estimated in 1–2 years (Cardenas et al., 2015). The effects of the 2004



tsunami were felt in a shallow sandy aquifer in Sri Lanka 1–1.5 years after the event (Vithanage et al., 2012). Salinization triggered by hurricanes Katrina and Rita persisted until 10 months after landfall in the groundwater system of coastal Louisiana (Van Biersel et al., 2007).

Vulnerability of coastal areas has been shown to decrease when recharge and hydraulic gradients increase (Yang et al., 2018). Topography itself has an impact on groundwater salinization. Saltwater accumulated in depressions increases the mass of infiltrated salt, whereas connected channels can promote more extensive salinization (Yu et al., 2016).

Many studies have focused on the hydrological, ecological, and geomorphological consequences of hurricanes, characterized by large storm surges (Fagherazzi, Anisfeld, et al., 2019; Fagherazzi, Nordio, et al., 2019; Fernandes et al., 2018; Gardner et al., 2002; Middleton, 2016). Recently, more attention has been paid about the effects of frequent and moderate storm events on coastal areas (Beebe et al., 2022; Wilson et al., 2011, 2015). However, the coupling between hydrological processes and coastal ecosystems has not been studied in detail. Beebe et al. (2022) estimated the consequences of a moderate storm on submarine groundwater discharge (SGD), with the formation of an anomalous seawater intrusion in the groundwater system. Wilson et al. (2015) determined the magnitude and the main processes controlling temporal variations in tidally driven SGD caused by moderate storm surges. However, most of these studies focus on a specific site, and do not provide a broad picture of the groundwater response at the regional scale in different coastal environments. Here, we sought to quantify the effect of more frequent events on coastal groundwater dynamics, extending the analysis at the regional scale in very different settings, and to discuss the consequences of moderate storm surge events on the coastal ecosystems. Between 11 and 14 October 2019, tropical storm Melissa hit the North Atlantic shoreline from North Carolina to Massachusetts. The storm developed over the western Atlantic Ocean, never made landfall, and evolved from subtropical storm to tropical storm on 12 October (Figure 1). Coastal water levels reached between 1 and 1.5 m above predicted levels (National Oceanic and Atmospheric Administration [NOAA]). An event of the magnitude of tropical storm Melissa has a return period (RP), or frequency, of about 1–2 years. Here we synthesized available groundwater level and specific conductivity measurements collected in the North Atlantic region between 9 and 15 October 2019, to assess the effects of a frequent, medium-intensity storm surge on coastal groundwater at a regional scale.

2. Methods and Data

Methods and data used are explained in Texts S1 and S2 in Supporting Information S1.

3. Results

3.1. Increase in Groundwater Level Correlates With Distance From the Shore and Ground Elevation

Groundwater data were analyzed in wells distributed along the North Atlantic coast from North Carolina to Massachusetts (Figure 1, Tables S1–S3, Text S1 in Supporting Information S1). The increase in groundwater level during the storm surge was significantly correlated with distance from the coast ($R^2 = 0.61$, p < 0.05) (Figure 2a) and ground surface elevation ($R^2 = 0.57$, p < 0.05) (Figure 2b). As expected, the effect of the storm on groundwater levels diminishes at greater distances from the coast and higher ground elevations. Storm effects on groundwater reached a minimum value of 5 cm at 4 km from the coast and at a ground surface elevation of 11 m on NAVD88 (Figure 2a). Groundwater increments could also be attributed to freshwater contributions from rainfall. Therefore, the analysis was repeated excluding stations that recorded high rainfall rates (Data). The relationship did not significantly change when these stations were excluded (Figures S1c and S1d, Texts S1 and S2 in Supporting Information S1). A multiple regression analysis considering both topographic elevation and distance from the coast as independent variables was significant (adjusted- $R^2 > 0.65$, p < 0.01) (Table S4, Text S2 in Supporting Information S1). The mean and maximum water specific conductivity of the groundwater were not significantly correlated to elevation, distance from the shore, or well depth ($R^2 < 0.3$, p > 0.05) (Figures S3, S4, and Text S2 in Supporting Information S1).



Geophysical Research Letters



Figure 1. Well locations along the North Atlantic coast. Melissa storm track between 11 and 14 October 2019, according to National Oceanic and Atmospheric Administration (a). Before 11 October the storm was classified as extratropical (dashed white line). The storm was classified as subtropical between 11 and 12 October (orange lines) and as tropical from 12 to 14 October (yellow lines). (b) Well distribution in a forested area in Hog Island Bay; (c) well distribution on a beach in Chincoteague Bay.

3.2. Groundwater System Recovery After Storm Surge Event

Groundwater level and specific conductivity were significantly different before and after the storm $(t(3.24) > t_{crit}(2.09), p < 0.05$ for groundwater level and $t(3.82) > t_{crit}(2.31), p < 0.05$ for groundwater specific conductivity), suggesting that the pre-storm conditions did not immediately re-establish (Figures 3, 4a, and 4b). The median value of groundwater level after the storm was 6.25 cm with respect to pre-storm values. The median specific conductivity before the storm was 13.74 mS/cm, in comparison to a median specific conductivity of 31.50 mS/cm after the storm (Figure 4b). Overall, conductivity values reached along the North Atlantic coast during the Melissa storm event are comparable to seawater conductivity values, ranging from 20 in Chesapeake Bay (NOAA buoy id: 8635750) and 39 mS/cm in Delaware coast (NOAA buoy id: 8557380). Variability in groundwater levels before the storm is significantly higher than after, likely because of differences in vegetation, soil, and land use. Once the storm surge occurred, groundwater levels and specific conductivity became similar across sites.

The groundwater recovery duration from the Melissa surge varied significantly by site ($t(2.78) > t_{crit}(2.31)$, p < 0.05). Our analysis suggests that groundwater levels recovered faster than groundwater specific conductivity. The median RT was 2.2 days for groundwater levels, in comparison to 20 days for specific conductivity (Figure 4c). Additionally, RT of groundwater specific conductivity was more variable across sites. In particular, recovery times for conductivity estimated in clay soil were much higher than those estimated in sandy soil. In the



Geophysical Research Letters



Figure 2. Increase in groundwater level during Melissa versus distance from the (a) coast and (b) ground elevation. Shaded areas represent 0.95 confidence interval. Flooding = sites flooded from above during Melissa, No-Flooding = sites not flooded from above.

clay forested area close to Hog Island Bay (VA) (Figure 1, Table S1 in Supporting Information S1), characterized by soil macropores and often wet, mean time to recover was 23 days. In the marshland in Leipsic (DE) (Figure 1, Table S1 in Supporting Information S1), characterized by clay soil, a smaller conductivity increase, due to the storm surge event, dampened by marshland vegetation friction, run out in a mean RT of around 4 days. In the sandy beach in Chincoteague Bay (DE) (Figure 1, Table S1 in Supporting Information S1), the estimated RT was around 17 hr on average. The RT of groundwater specific conductivity had an interquartile range of 26 days, nine times larger than the interquartile range of groundwater level RT (Figure 4c). Recovery time for both groundwater level and conductivity were found to be not correlated with distance from the ocean, ground elevation, and well depths (not shown). Moreover, recovery timescale of groundwater level was not clearly correlated to sediment type at the wells' screen, suggesting a possible strong influence of external inputs, local morphology, and vegetation (Figure S5 in Supporting Information S1).



Figure 3. (a) Groundwater level and (b) specific conductivity data in Hog Island forested areas H5, M1, and L1 and in the deep well CT along the Chincoteague beach. Specific conductivity thresholds for *Pinus taeda* forests under controlled conditions are identified according to Woods et al. (2020). Photosynthetic activity of *P. taeda* is optimal when specific conductivity conditions are below 5 ppt in unsaturated soil conditions (Poulter et al., 2008; Woods et al., 2020).



Geophysical Research Letters



Figure 4. (a) Peak and after storm groundwater levels compared to pre-storm values, (b) groundwater specific conductivity values before and after a storm surge event. (c) Recovery timescale for groundwater level and specific conductivity.

Ratios between groundwater recovery timescale and RP for the Melissa storm were compared to similar ratios derived from the literature for other storms that affected coastal sites around the world (Figure 5). Ratios represent the percent of time during which the system is characterized by an anomalous conductivity and accounts for both the frequency and magnitude of the events. The different hydrological settings of the study sites justify the ratio variability. Despite the low storm surge, Melissa ratios were similar to ratios representing very energetic hurricanes of category 3 and 4. This is because the RP of Melissa is much lower than a hurricane, thus increasing the percent of time with salinization. During Melissa, the ratio reached a value of 0.054 in clay soils, the highest values calculated in the analysis. Here, clay soil and the presence of soil macropores, mostly created by uprooting of dead trees and dense understory vegetation, encourage water retention and water infiltration during storm events, significantly increasing groundwater conductivity and making recovery times longer. A lower ratio of 0.022 estimated in a marsh in Leipsic (DE clay soil) (Figure 1, Table S1 in Supporting Information S1) was reached in a sandy beach in Chincoteague Bay (DE) (Figure 1, Table S1 in Supporting Information S1), due to the weaker ability of the soil to retain water.



Figure 5. Ratios between recovery time and return period for different storm events. Data were derived from the literature (Anderson, 2002; Cardenas et al., 2015; Hedgespeth et al., 2021; Keim et al., 2007; Kiflai et al., 2020; McDowell et al., 1996; Sawyer et al., 2014; Terry & Falkland, 2010; Van Biersel et al., 2007; Vithanage et al., 2012; Wachnicka et al., 2020; Williams, 1993).

4. Discussion

The medium-intensity storm surge (RP ~1-2 years) of Tropical Storm Melissa affected coastal groundwater levels and specific conductivity at a regional scale along the eastern shore of the United States. The groundwater level increase due to the storm surge is significantly larger in wells closer to the ocean and at lower elevation. Nine wells at four sites, reaching a maximum distance of 500 m from the coast, comprise a representative sample of coastal environments on the northeastern seaboard. In our analysis we quantify the overall effect of a moderate storm surge event on coastal groundwater systems located in different landscapes. For instance, similar or smaller events can be significantly felt in barrier islands, where run-up mechanisms encourage an increase in groundwater level and conductivity. The same events can be slightly or no felt in inland marshland areas, where the water signal reaching the wells is dampened. Tidal pumping and wave setup are the main driving forces of groundwater flow through the beach and they tend to exacerbate the storm surge effects on sloping surfaces (Evans & Wilson, 2017; Nielsen, 1999). On the other hand, during its flow from the ocean to a coastal forest bordering a marshland, the water energy dissipates due to marsh vegetation friction, and the storm surge effect is reduced (Leonardi et al., 2018; Nordio and Fagherazzi, 2022; Stark et al., 2015). While we are able to draw general conclusions from this limited sample, these results highlight the value of additional intensive groundwater monitoring in the coastal zone, to capture the dynamics of SLR and coastal saltwater intrusion. Since medium intensity storm events have been occurring more frequently in the last century (Knutson et al., 2010; Landsea et al., 2010), they could change groundwater characteristics and consequently undermine ecological communities.

Our data suggest that a medium-intensity storm surge can negatively affect groundwater dynamics, preventing a complete recovery once the storm surge stops. Soil needs time to drain to restore the levels present before the storm surge. Drainage can require between 1 and 10 days in the absence of similar events. As the soil starts to drain, the salt concentration in the groundwater tends to decrease. The recovery timescale in shallow aquifers is around 10 times higher for specific conductivity than for groundwater level. Inter-storm arrival times (Khaertidova & Longobardi, 2013) and hydrogeological characteristics control RT (Knutson et al., 2010; Terry & Falkland, 2010; Yang et al., 2018). Evapotranspiration rate can also accelerate the groundwater level recovery (Gardner et al., 2002). At the same time, storm surge salinization is exacerbated by both evaporation and evapotranspiration processes. In intertidal coastal areas, salinity values are often higher than those recorded in seawater (Geng & Boufadel, 2015; Geng et al., 2016, 2021; Rajmohan et al., 2021). This especially occurs in the upper intertidal zone, that is less affected by tides and waves inundation, so that the new water can dilute pore water salinity (Geng et al., 2016). When meteorological conditions are favorable to high evaporation (i.e., high temperature and low humidity) capillary fringe increase soil moisture in the upper soil layer, encouraging water evaporation and consequently salt concentration (Geng & Boufadel, 2015; Geng et al., 2016, 2021). Groundwater modeling conducted on sandy beaches suggests a doubling in salinity in the intertidal zone when evaporation is considered (Geng & Boufadel, 2015). Geng and Boufadel (2015), showed that after maximum salinity values are reached during a spring tide, salinity decreased slowly due to the brackish water surrounding the beach site. After storm surge events, salinity values are higher, and evapotranspiration likely increases salinization with deleterious effects for non-salt tolerant ecosystems. The estimated time to recover from a medium-intensity storm surge event is shorter than the RT after a hurricane (Cardenas et al., 2015; Vithanage et al., 2012). However, hurricanes are less frequent, and therefore the relative time during which salinization occurs is shorter.

Specific conductivity dynamics follow the groundwater level and an upward salt flux toward the root zone can trigger changes in plant health (Mohamed et al., 2000; Munns & Tester, 2008). A recovery timescale of months for groundwater specific conductivity can be crucial for salt-intolerant vegetation, particularly if it occurs with a RP of only 1 or 2 years. In this scenario, along forested areas close to marshland, mature trees can defoliate or die (Fagherazzi, Anisfeld, et al., 2019; Fagherazzi, Nordio, et al., 2019). *Pinus taeda* dominates maritime forests in the southern portion of our study region (Hog Island and Chesapeake Bay sites). This tree is moderately flood-tolerant (Pezeshki, 1992) and able to establish and survive in specific conductivity conditions up to 5 ppt (~8 mS/cm at 20°C) (Poulter et al., 2008; Woods et al., 2020). Seedlings are more sensitive to specific conductivity increase while mature trees can show greater salt tolerance (Kirwan et al., 2007; Poulter et al., 2008). Woods et al. (2020) observed that, under controlled conditions in a growth chamber, *Pinus taeda* germination was unaffected at up to 10 ppt (~17 mS/cm at 20°C) salinity and germination was reduced by half when salinity was 20 ppt (~32 mS/cm at 20°C). The groundwater specific conductivity levels reached during the Melissa storm surge were between two and five times higher than the tolerated conductivity levels. These high salinity levels



can affect photosynthetic dynamics, stomatal conductance, and biomass production (Pezeshki, 1992). Stress due to inundation can compound the stress of saltwater exposure (Pezeshki, 1992; Poulter et al., 2008). Pitch pine (*Pinus rigida*), black oak (*Quercus velutina*), and white oak (*Quercus alba*) dominate forests at the Wellfleet site (Hall et al., 2002; Smith et al., 2011). Pitch pines are quite intolerant to salt spray (Griffiths & Orians, 2004) and succumb to total immersion in two to 4 weeks (Craine & Orians, 2006). Oaks, however, can survive but only at conductivity levels up to 4 mS/cm (Kotuby-Amacher et al., 2000). In farmland sites, salinity levels tolerated by the crops are generally up to 2 ppt (~4 mS/cm at 20°C) (Tully et al., 2019), much lower than salinity levels reached during Melissa. Storm surge events not only directly affect the survival of native species, but also encourage encroachment of more flooding and salt tolerant plants that establish in the disturbed ecosystem (Noto & Shurin, 2017).

According to our analysis, the estimated groundwater recovery times are sufficiently lower than the RP of Melissa so we can suppose that, in absence of external competitors, once original conditions are re-established, vegetation regeneration restarts. However, SLR and global warming will decrease the RP of these storms in the near future, and a subtropical storm event like Melissa will have more drastic consequences. For example, with a sea level increase of 50 cm, the magnitude of a storm surge like Melissa could become comparable to storm surges that today have a return time between 5 and 14 years (Figure S6 in Supporting Information S1). Therefore, moderate but frequent storm events could affect ecosystems survival, frequently changing hydrological conditions and consequently stressing photosynthetic activity of native species (Budke et al., 2008; Vreugdenhil et al., 2006). Although major hurricanes and storms have catastrophic consequences on society and economy, moderate but frequent storm surge events also contribute to ecological change.

Our analysis indicates that the groundwater level and specific conductivity are similar across the studied wells after Melissa, suggesting more homogeneous groundwater post-storm conditions. This mainly occurs in shallow aquifers. Hydrological variability can be a crucial driver of biodiversity at most scales of analysis (Konar et al., 2013). Therefore, this homogenizing phenomenon might drastically reduce biodiversity and consequently affect ecosystem functioning (Konar et al., 2013). At some coastal locations, SLR and frequent tropical storms are already reshaping the landscape with an irreversible impact on biodiversity (Allen & Lendemer, 2016; Burkett et al., 2008), and consequently affect socio-economic development (Midgley, 2012; Sylvain & Wall, 2011). Here we put forward the hypothesis that homogenization of conductivity and groundwater levels driven by moderate storms can be partly responsible for loss of biodiversity.

The United States Geological Survey (USGS) stations used herein do not include specific conductivity measurements. The paucity of available conductivity data hampered our analysis on the extent of groundwater salinization due to Tropical Storm Melissa. Conductivity monitoring across a broad swath of the coastal zone is imperative to detect how climate change affects coastal groundwater resources. Our study is unique because in the past, salinization and flooding due to storm surge events have been investigated at the continental scale using only numerical models (Guimond & Michael, 2021; Knutson et al., 2010; Paldor & Michael, 2021; Yang et al., 2018). This research illustrates the value of field studies of saltwater intrusion and highlights a broader need for large-scale data sets.

5. Conclusions

Our analysis demonstrates that frequent storm surges (RP \sim 1–2 years) affect groundwater at a regional scale. Groundwater level and specific conductivity increased after Tropical Storm Melissa up to 4 km inland. The change in groundwater level decreased with elevation and distance from the coast. Recovery time of groundwater specific conductivity was 10 times greater than the RT of water level in shallow wells. A frequency-magnitude analysis indicates that the percent of time with salinization after Melissa is of the same order if not higher than the relative salinization periods of energetic hurricanes and tsunamis, highlighting the potential damage of moderate but frequent storms on coastal ecosystems.

Long high-conductivity periods can trigger forest dieback and favor the encroachment of new vegetation species. Groundwater level and specific conductivity after the storm surge were more uniform. This homogenizing process can trigger loss in biodiversity, encouraging the establishment of monotonic halophyte vegetation. Large-scale efforts to monitor groundwater conductivity in the coastal zone can help to characterize the effect of ever-more frequent storms on groundwater and related ecosystems to inform resource management.



Acknowledgments

Data Availability Statement

Part of the data supporting findings of this research are openly available in the Long Term Ecological Research-Virginia Coast Reserve (LTER-VCR) at https://doi.org/10.6073/pasta/942a5a981e6e986c-5fa1a9a9cd2eb8b7, in the National Oceanic and Atmospheric Association (NOAA) repository at https://www.noaa.gov/ and in the United States Geological Survey (USGS) repository at https://doi.org/10.5066/ P9XQ27F5 and https://doi.org/10.5066/F7P55KJN.

References

- Allen, J. L., & Lendemer, J. C. (2016). Quantifying the impacts of sea-level rise on coastal biodiversity: A case study on lichens in the mid-Atlantic coast of eastern North America. *Biological Conservation*, 202, 119–126. https://doi.org/10.1016/j.biocon.2016.08.031
- Anderson, W. P., Jr. (2002). Aquifer salinization from storm overwash. Journal of Coastal Research, 18(3), 413–420. Retrieved from http://www. jstor.org/stable/4299090
- Antonellini, M., & Mollema, P. N. (2010). Impact of groundwater salinity on vegetation species richness in the coastal pine forests and wetlands of Ravenna, Italy. *Ecological Engineering*, 36(9), 1201–1211. https://doi.org/10.1016/j.ecoleng.2009.12.007
- Barlow, P. M., & Reichard, E. G. (2009). Saltwater intrusion in coastal regions of North America. *Hydrogeology Journal*, 18(1), 247–260. https://doi.org/10.1007/s10040-009-0514-3
- Beebe, D. A., Huettemann, M. B., Webb, B. M., & Jackson, W. T., Jr. (2022). Atmospheric groundwater forcing of a subterranean estuary: A seasonal seawater recirculation process. *Geophysical Research Letters*, 49(7), e2021GL096154. https://doi.org/10.1029/2021gl096154
- Budke, J. C., Jarenkow, J. A., & de Oliveira-Filho, A. T. (2008). Tree community features of two stands of riverine forest under different flooding regimes in Southern Brazil. Flora-Morphology, Distribution, Functional Ecology of Plants, 203(2), 162–174. https://doi.org/10.1016/j. flora.2007.03.001
- Burkett, V. R., Nicholls, R. J., Fernandez, L., & Woodroffe, C. D. (2008). Climate change impacts on coastal biodiversity.
- Cardenas, M. B., Bennett, P. C., Zamora, P. B., Befus, K. M., Rodolfo, R. S., Cabria, H. B., & Lapus, M. R. (2015). Devastation of aquifers from tsunami-like storm surge by Supertyphoon Haiyan. *Geophysical Research Letters*, 42(8), 2844–2851. https://doi.org/10.1002/2015gl063418 Craine, S. I., & Orians, C. M. (2006). Effects of flooding on pitch pine (*Pinus rigida* Mill.) growth and survivorship. *Journal of the Torrey Botan*-
- ical Society, 133(2), 289–296. https://doi.org/10.3159/1095-5674(2006)133[289:eofopp]2.0.co;2
- Evans, T. B., & Wilson, A. M. (2017). Submarine groundwater discharge and solute transport under a transgressive barrier island. Journal of Hydrology, 547, 97–110. https://doi.org/10.1016/j.jhydrol.2017.01.028
- Fagherazzi, S., Anisfeld, S. C., Blum, L. K., Long, E. V., Feagin, R. A., Fernandes, A., et al. (2019). Sea level rise and the dynamics of the marsh-upland boundary. *Frontiers in Environmental Science*, 7, 25. https://doi.org/10.3389/fenvs.2019.00025
- Fagherazzi, S., Nordio, G., Munz, K., Catucci, D., & Kearney, W. S. (2019). Variations in persistence and regenerative zones in coastal forests triggered by sea level rise and storms. *Remote Sensing*, 11(17), 2019. https://doi.org/10.3390/rs11172019
- Fernandes, A., Rollinson, C. R., Kearney, W. S., Dietze, M. C., & Fagherazzi, S. (2018). Declining radial growth response of coastal forests to hurricanes and nor'easters. *Journal of Geophysical Research: Biogeosciences*, 123(3), 832–849. https://doi.org/10.1002/2017jg004125
- Gardner, L. R., Reeves, H. W., & Thibodeau, P. M. (2002). Groundwater dynamics along forest-marsh transects in a southeastern salt marsh, USA: Description, interpretation and challenges for numerical modeling. *Wetlands Ecology and Management*, 10(2), 143–157. https://doi. org/10.1023/a:1016571909992
- Geng, X., & Boufadel, M. C. (2015). Impacts of evaporation on subsurface flow and salt accumulation in a tidally influenced beach. Water Resources Research, 51(7), 5547–5565. https://doi.org/10.1002/2015wr016886
- Geng, X., Boufadel, M. C., & Jackson, N. L. (2016). Evidence of salt accumulation in beach intertidal zone due to evaporation. *Scientific Reports*, 6(1), 1–5. https://doi.org/10.1038/srep31486
- Geng, X., Khalil, C. A., Prince, R. C., Lee, K., An, C., & Boufadel, M. C. (2021). Hypersaline pore water in Gulf of Mexico beaches prevented efficient biodegradation of Deepwater Horizon beached oil. *Environmental Science & Technology*, 55(20), 13792–13801. https://doi.org/10.1021/acs.est.1c02760
- Griffiths, M. E., & Orians, C. M. (2004). Salt spray effects on forest succession in rare coastal sandplain heathlands: Evidence from field surveys and *Pinus rigida* transplant experiments. *Journal of the Torrey Botanical Society*, 131(1), 23–31. https://doi.org/10.2307/4126925
- Guimond, J. A., & Michael, H. A. (2021). Effects of marsh migration on flooding, saltwater intrusion, and crop yield in coastal agricultural land subject to storm surge inundation. Water Resources Research, 57(2), e2020WR028326. https://doi.org/10.1029/2020wr028326
- Hall, B., Motzkin, G., Foster, D. R., Syfert, M., & Burk, J. (2002). Three hundred years of forest and land-use change in Massachusetts, USA. Journal of Biogeography, 29(10-11), 1319–1335. https://doi.org/10.1046/j.1365-2699.2002.00790.x
- Hallett, R., Johnson, M. L., & Sonti, N. F. (2018). Assessing the tree health impacts of salt water flooding in coastal cities: A case study in New York City. *Landscape and Urban Planning*, *177*, 171–177. https://doi.org/10.1016/j.landurbplan.2018.05.004
- Khaertidova, E., & Longobardi, A. (2013). Analysis of inter-storm period soil moisture dynamics. Procedia Environmental Sciences, 19, 208–216. https://doi.org/10.1016/j.proenv.2013.06.023
- Kirwan, M. L., & Gedan, K. B. (2019). Sea-level driven land conversion and the formation of ghost forests. *Nature Climate Change*, 9(6), 450–457. https://doi.org/10.1038/s41558-019-0488-7
- Kirwan, M. L., Kirwan, J. L., & Copenheaver, C. A. (2007). Dynamics of an estuarine forest and its response to rising sea level. Journal of Coastal Research, 23(2), 457–463. https://doi.org/10.2112/04-0211.1
- Knutson, T. R., McBride, J. L., Chan, J., Emanuel, K., Holland, G., Landsea, C., et al. (2010). Tropical cyclones and climate change. Nature Geoscience, 3(3), 157–163. https://doi.org/10.1038/ngeo779
- Konar, M., Todd, M. J., Muneepeerakul, R., Rinaldo, A., & Rodriguez-Iturbe, I. (2013). Hydrology as a driver of biodiversity: Controls on carrying capacity, niche formation, and dispersal. Advances in Water Resources, 51, 317–325. https://doi.org/10.1016/j.advwatres.2012.02.009
- Kotuby-Amacher, J., Koenig, R., & Kitchen, B. (2000). Salinity and plant tolerance. Electronic Publication AG-SO-03, Utah State University Extension.
- Landsea, C. W., Vecchi, G. A., Bengtsson, L., & Knutson, T. R. (2010). Impact of duration thresholds on Atlantic tropical cyclone counts. *Journal of Climate*, 23(10), 2508–2519. https://doi.org/10.1175/2009jcli3034.1
- Leonardi, N., Carnacina, I., Donatelli, C., Ganju, N. K., Plater, A. J., Schuerch, M., & Temmerman, S. (2018). Dynamic interactions between coastal storms and salt marshes: A review. *Geomorphology*, *301*, 92–107. https://doi.org/10.1016/j.geomorph.2017.11.001

This research was funded by the USA National Science Foundation awards 1832221 (VCR LTER), 2224608 (PIE LTER), and 2012322 (Coastal Critical Zone [CZN]). Any use of trade, firm, or product names is for descriptive purposes only and does not imply endorsement by the U.S. Government. Joel Carr acknowledges support from the USGS Climate Research and Development Program and Ecosystems Mission Area.



- Middleton, B. A. (2016). Differences in impacts of Hurricane Sandy on freshwater swamps on the Delmarva Peninsula, Mid-Atlantic Coast, USA. *Ecological Engineering*, 87, 62–70. https://doi.org/10.1016/j.ecoleng.2015.11.035
- Midgley, G. F. (2012). Biodiversity and ecosystem function. Science, 335(6065), 174-175. https://doi.org/10.1126/science.1217245
- Mohamed, A. A., Sasaki, T., & Watanabe, K. (2000). Solute transport through unsaturated soil due to evaporation. Journal of Environmental Engineering, 126(9), 842–848. https://doi.org/10.1061/(asce)0733-9372(2000)126:9(842)
- Munns, R., & Tester, M. (2008). Mechanisms of salinity tolerance. Annual Review of Plant Biology, 59(1), 651–681. https://doi.org/10.1146/ annurev.arplant.59.032607.092911
- Nielsen, P. (1999). Groundwater dynamics and salinity in coastal barriers. Journal of Coastal Research, 15(3), 732–740. Retrieved from https://www.jstor.org/stable/4298987
- Nordio, G., & Fagherazzi, S. (2022). Storm surge and tidal dissipation in deltaic wetlands bordering a main channel. Journal of Geophysical Research: Oceans, 127(3), e2021JC017655. https://doi.org/10.1029/2021jc017655
- Noto, A. E., & Shurin, J. B. (2017). Early stages of sea-level rise lead to decreased salt marsh plant diversity through stronger competition in Mediterranean-climate marshes. *PLoS One*, 12(1), e0169056. https://doi.org/10.1371/journal.pone.0169056
- Paldor, A., & Michael, H. A. (2021). Storm surges cause simultaneous salinization and freshening of coastal aquifers, exacerbated by climate change. Water Resources Research, 57(5), e2020WR029213. https://doi.org/10.1029/2020wr029213

Pezeshki, S. (1992). Response of Pinus taeda L to soil flooding and salinity. Annales Des Sciences Forestières, 49(2), 149–159. https://doi. org/10.1051/forest:19920205

- Poulter, B., Christensen, N. L., & Qian, S. S. (2008). Tolerance of *Pinus taeda* and *Pinus serotina* to low salinity and flooding: Implications for equilibrium vegetation dynamics. *Journal of Vegetation Science*, 19(1), 15–22. https://doi.org/10.3170/2007-8-18410
- Rajmohan, N., Masoud, M. H., & Niyazi, B. A. (2021). Impact of evaporation on groundwater salinity in the arid coastal aquifer, Western Saudi Arabia. Catena, 196, 104864. https://doi.org/10.1016/j.catena.2020.104864
- Schieder, N. W., & Kirwan, M. L. (2019). Sea-level driven acceleration in coastal forest retreat. Geology, 47(12), 1151–1155. https://doi. org/10.1130/g46607.1
- Smith, S. M., Decker, V., & Phillips, C. (2011). Coastal forest monitoring protocol, Cape Cod National Seashore. In Natural Resource Report NPS/CACO/NRR—2011/388. National Park Service.
- Stark, J., Van Oyen, T., Meire, P., & Temmerman, S. (2015). Observations of tidal and storm surge attenuation in a large tidal marsh. Limnology and Oceanography, 60(4), 1371–1381. https://doi.org/10.1002/lno.10104
- Sylvain, Z. A., & Wall, D. H. (2011). Linking soil biodiversity and vegetation: Implications for a changing planet. American Journal of Botany, 98(3), 517–527. https://doi.org/10.3732/ajb.1000305
- Terry, J. P., & Falkland, A. C. (2010). Responses of atoll freshwater lenses to storm-surge overwash in the Northern Cook Islands. Hydrogeology Journal, 18(3), 749–759. https://doi.org/10.1007/s10040-009-0544-x
- Tully, K., Gedan, K., Epanchin-Niell, R., Strong, A., Bernhardt, E. S., BenDor, T., et al. (2019). The invisible flood: The chemistry, ecology, and social implications of coastal saltwater intrusion. *BioScience*, 69(5), 368–378. https://doi.org/10.1093/biosci/biz027
- Van Biersel, T. P., Carlson, D. A., & Milner, L. R. (2007). Impact of hurricanes storm surges on the groundwater resources. *Environmental Geology*, 53(4), 813–826. https://doi.org/10.1007/s00254-007-0694-x
- Vithanage, M., Engesgaard, P., Villholth, K. G., & Jensen, K. H. (2012). The effects of the 2004 tsunami on a coastal aquifer in Sri Lanka. Groundwater, 50(5), 704–714. https://doi.org/10.1111/j.1745-6584.2011.00893.x
- Vreugdenhil, S. J., Kramer, K., & Pelsma, T. (2006). Effects of flooding duration, -frequency and-depth on the presence of saplings of six woody species in north-west Europe. Forest Ecology and Management, 236(1), 47–55. https://doi.org/10.1016/j.foreco.2006.08.329
- Williams, K., Ewel, K. C., Stumpf, R. P., Putz, F. E., & Workman, T. W. (1999). Sea-level rise and coastal forest retreat on the west coast of Florida, USA. *Ecology*, 80(6), 2045–2063. https://doi.org/10.1890/0012-9658(1999)080[2045:slracf]2.0.co;2
- Wilson, A. M., Evans, T. B., Moore, W. S., Schutte, C. A., & Joye, S. B. (2015). What time scales are important for monitoring tidally influenced submarine groundwater discharge? Insights from a salt marsh. Water Resources Research, 51(6), 4198–4207. https://doi. org/10.1002/2014wr015984
- Wilson, A. M., Moore, W. S., Joye, S. B., Anderson, J. L., & Schutte, C. A. (2011). Storm-driven groundwater flow in a salt marsh. Water Resources Research, 47(2), W02535. https://doi.org/10.1029/2010wr009496
- Woods, N. N., Swall, J. L., & Zinnert, J. C. (2020). Soil salinity impacts future community composition of coastal forests. Wetlands, 40(5), 1495–1503. https://doi.org/10.1007/s13157-020-01304-6
- Xiao, H., Wang, D., Medeiros, S. C., Bilskie, M. V., Hagen, S. C., & Hall, C. R. (2019). Exploration of the effects of storm surge on the extent of saltwater intrusion into the surficial aquifer in coastal east-central Florida (USA). Science of the Total Environment, 648, 1002–1017. https:// doi.org/10.1016/j.scitotenv.2018.08.199
- Yang, J., Zhang, H., Yu, X., Graf, T., & Michael, H. A. (2018). Impact of hydrogeological factors on groundwater salinization due to ocean-surge inundation. Advances in Water Resources, 111, 423–434. https://doi.org/10.1016/j.advwatres.2017.11.017
- Yu, X., Yang, J., Graf, T., Koneshloo, M., O'Neal, M. A., & Michael, H. A. (2016). Impact of topography on groundwater salinization due to ocean surge inundation. *Water Resources Research*, 52(8), 5794–5812. https://doi.org/10.1002/2016wr018814

References From the Supporting Information

- Beven, K., & Germann, P. (1982). Macropores and water flow in soils. Water Resources Research, 18(5), 1311–1325. https://doi.org/10.1029/ wr018i005p01311
- Bilardi, S., Ielo, D., & Moraci, N. (2020). Predicting the saturated hydraulic conductivity of clayey soils and clayey or silty sands. *Geosciences*, 10(10), 393. https://doi.org/10.3390/geosciences10100393
- Bouma, J. (1982). Measuring the hydraulic conductivity of soil horizons with continuous macropores. Soil Science Society of America Journal, 46(2), 438–441. https://doi.org/10.2136/sssaj1982.03615995004600020047x
- Carr, J. A., & Guntenspergen, G. R. (2020). Water levels (November 11 2016 through November 11 2017) for four wells and light intensity data (October 1 2015 through September 2019): From marsh to upland forest, for moneystump marsh, blackwater National Wildlife Refuge, Maryland. U.S. Geological Survey data release. https://doi.org/10.5066/P9XQ27F5
- Colloff, M. J., Pullen, K. R., & Cunningham, S. A. (2010). Restoration of an ecosystem function to revegetation communities: The role of invertebrate macropores in enhancing soil water infiltration. *Restoration Ecology*, 18, 65–72. https://doi.org/10.1111/j.1526-100x.2010.00667.x
- Fagherazzi, S., & Nordio, G. (2022). Groundwater, soil moisture, light and weather data in Brownsville forest, 2019-2022 ver 4. Environmental Data Initiative. https://doi.org/10.6073/pasta/942a5a981e6e986c5fa1a9a9cd2eb8b7



- Hedgespeth, M. L., McCord, J. P., Phillips, K. A., Strynar, M. J., Shea, D., & Nichols, E. G. (2021). Suspect-screening analysis of a coastal watershed before and after Hurricane Florence using high-resolution mass spectrometry. *Science of the Total Environment*, 782, 146862. https://doi. org/10.1016/j.scitotenv.2021.146862
- Keim, B. D., Muller, R. A., & Stone, G. W. (2007). Spatiotemporal patterns and return periods of tropical storm and hurricane strikes from Texas to Maine. Journal of Climate, 20(14), 3498–3509. https://doi.org/10.1175/jcli4187.1
- Kiflai, M. E., Whitman, D., Ogurcak, D. E., & Ross, M. (2020). The effect of Hurricane Irma storm surge on the freshwater lens in Big Pine Key, Florida using electrical resistivity tomography. *Estuaries and Coasts*, 43(5), 1032–1044. https://doi.org/10.1007/s12237-019-00666-3
- McDowell, W. H., McSwiney, C. P., & Bowden, W. B. (1996). Effects of hurricane disturbance on groundwater chemistry and riparian function in a tropical rain forest. *Biotropica*, 28(4), 577–584. https://doi.org/10.2307/2389098
- Nordio, G., & Fagherazzi, S. (2022). Groundwater, soil moisture, light and weather data collected in a coastal forest bordering a salt marsh in the Delmarva Peninsula (VA). *Data in Brief*, 45, 108584. https://doi.org/10.1016/j.dib.2022.108584
- Pittman, F., Mohammed, A., & Cey, E. (2020). Effects of antecedent moisture and macroporosity on infiltration and water flow in frozen soil. *Hydrological Processes*, 34(3), 795–809. https://doi.org/10.1002/hyp.13629
- Sawyer, A. H., Kaplan, L. A., Lazareva, O., & Michael, H. A. (2014). Hydrologic dynamics and geochemical responses within a floodplain aquifer and hyporheic zone during Hurricane Sandy. Water Resources Research, 50(6), 4877–4892. https://doi.org/10.1002/2013wr015101
- U.S. Geological Survey. (2016). National water information system data available on the world wide web. (USGS Water Data for the Nation). Retrieved from http://waterdata.usgs.gov/nwis/
- Wachnicka, A., Browder, J., Jackson, T., Louda, W., Kelble, C., Abdelrahman, O., et al. (2020). Hurricane Irma's impact on water quality and phytoplankton communities in Biscayne Bay (Florida, USA). *Estuaries and Coasts*, 43(5), 1217–1234. https://doi.org/10.1007/s12237-019-00592-4
- Williams, T. M. (1993). Salt water movement within the water table aquifer following Hurricane Hugo. In *Proceedings of the Seventh biennial southern silvicultural research conference* (pp. 177–183). USDA Forest Service. Southern Forest Experiment Station.